Author's response to reviews

Title: Left ventricular diastolic dysfunction of the cardiac surgery patient; a point of view for the Cardiac Surgeon and Anesthesiologist

Authors:

Efstratios E Apostolakis (stratisapostolakis@yahoo.gr)
Haralabos Parissis (hparissis@yahoo.co.uk)
Nikolaos G Baikoussis (ngbaik@yahoo.com)
Stavros N Siminelakis (ngbaik@yahoo.com)
Georgios S Papadopoulos (ngbaik@yahoo.com)

Version: 3 Date: 24 October 2009

Author's response to reviews: see over
Patras 24/10/2009

Editor in Chief
Journal of cardiothoracic surgery

Cover letter

Dear Sir/ Madame
We herewith submit for publication in journal of cardiothoracic surgery our review paper entitled: Left ventricular diastolic dysfunction of the cardiac surgery patient; a point of view for the Cardiac Surgeon and Anesthesiologist
Efstratios Apostolakis, Haralabos Parissis, Nikolaos G. Baikoussis, Stavros N. Siminelakis, Georgios Papadopoulos

The paper is prepared following the instructions of journal of cardiothoracic surgery on line. A revision was done. References order, figure 1 and its legend was added.

Sincerely yours
N.G. Baikoussis MD